Author's response to reviews

Title: Non-invasive pulse wave analysis for monitoring the cardiovascular effects of CO2 pneumoperitoneum during laparoscopic cholecystectomy - a prospective case-series study

Authors:

Péter Sárkány (sarkanypeter@hotmail.com)
Szabolcs Lengyel (szabolcs.lengyel@med.unideb.hu)
Réka Nemes (reka.nemes.11@gmail.com)
Livia Orosz (orly@yahoo.com)
Dénes Páll (pall@belklinika.com)
Csilla Molnár (csmolnar@med.unideb.hu)
Bela Fulesdi (fulesdi@med.unideb.hu)

Version: 5
Date: 20 August 2014

Author's response to reviews:

As suggested by the editor, to the methods section we added the sentence as follows: "A written informed consent to participate was obtained from all patients included in the study."

No other modifications were indicated.

Thank you for your editing efforts.